Author's response to reviews

Title: Loss of heterozygosity: what is it good for?

Authors:

Georgina L Ryland (georgina.ryland@petermac.org)
Maria A Doyle (maria.doyle@petermac.org)
David Goode (david.goode@petermac.org)
Samantha E Boyle (samantha.boyle@petermac.org)
David YH Choong (david.choong@petermac.org)
Simone M Rowley (simone.rowley@petermac.org)
Jason Li (jason.li@petermac.org)
David DL Bowtell (david.bowtell@petermac.org)
Richard W Tothill (richard.tothill@petermac.org)
Ian G Campbell (ian.campbell@petermac.org)
Kylie L Gorringe (kylie.gorringe@petermac.org)

Version: 5
Date: 8 July 2015

Author's response to reviews:

Response to second Editorial request.

1. Include keywords after the Abstract
   Done.

2. Clarify in the Authors contribution Section if that all authors read and approved the final manuscript.
   Done.

3. Upload the supplementary files in the correct order (currently supplementary table 4 appears before tables 1-3)
   Done.